

# Air traffic energy efficiency differs from place to place: analysis of historical trends by geographical zones using a macro-level methodology

Benoit Chèze, Pascal Gastineau, Julien Chevallier

### ▶ To cite this version:

Benoit Chèze, Pascal Gastineau, Julien Chevallier. Air traffic energy efficiency differs from place to place: analysis of historical trends by geographical zones using a macro-level methodology. 2011. hal-02474644

# HAL Id: hal-02474644 https://ifp.hal.science/hal-02474644v1

Preprint submitted on 9 Mar 2020

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.





Air traffic energy efficiency differs from place to place: analysis of historical trends by geographical zones using a macro-level methodology

> Benoît CHÈZE Pascal GASTINEAU Julien CHEVALLIER

Décembre 2011

Les cahiers de l'économie - n° 86

## Série Recherche

<u>benoit.cheze@ifpen.fr</u> <u>pascal.gastineau@ifsttar.fr</u> <u>chevallier@dauphine.fr</u>

La collection "Les cahiers de l'économie" a pour objectif de présenter des travaux réalisés à IFP Energies nouvelles et à IFP School, travaux de recherche ou notes de synthèse en économie, finance et gestion. La forme peut être encore provisoire, afin de susciter des échanges de points de vue sur les sujets abordés. Les opinions émises dans les textes publiés dans cette collection doivent être considérées comme propres à leurs auteurs et ne reflètent pas nécessairement le point de vue d' IFP Energies nouvelles ou d' IFP School.

Pour toute information sur le contenu, prière de contacter directement l'auteur. Pour toute information complémentaire, prière de contacter le Centre Économie et Gestion: Tél +33 1 47 52 72 27

The views expressed in this paper are those of the authors and do not imply endorsement by the IFP Energies Nouvelles or the IFP School. Neither these institutions nor the authors accept any liability for loss or damage incurred as a result of the use of or reliance on the content of this publication.

# Air traffic energy efficiency differs from place to place: analysis of historical trends by geographical zones using a macro-level methodology<sup>\*</sup>

Benoît Chèze<sup>† ‡ §</sup>, Pascal Gastineau<sup>¶ ‡</sup>, and Julien Chevallier<sup>|| ‡</sup>

December 14, 2011

#### Abstract

This article analyses energy efficiency coefficients and their evolution in the air transport sector. The proposed 'macro-level' methodology allows obtaining energy efficiency coefficients and their growth rates (corresponding to the evolution of energy gains) from 1983 to 2006 for eight distinct geographical regions and at the world level. During the whole period, energy efficiency improvements have been equal to 2.88% per year at the world level, with strong differences between regions. Moreover, our results indicate that domestic air travels are less energy efficient (i.e. more carbon intensive) than international air travels. This result applies in all regions.

**Keywords:** Air Transport, Technological Innovation, Energy Efficiency, Carbon Intensity, Macro-Level Methodology.

JEL Classification Numbers: L93, Q54, Q55.

<sup>\*</sup>The research leading to these results has received funding from the European Community's Seventh Framework Programme (FP7/2007-2013) under grant agreement ACP7-GA-213266 (Alfa–Bird). The authors are grateful to Laurie Starck (IFP Energies nouvelles (IFPEN)) for her valuable help. The authors also wish to thank for their helpful comments Laurent Meunier (ADEME) and from the IFPEN: Emilie Bertout, Nicolas Jeuland, Pierre Marion, Axel Pierru, J. Sabathier, Valérie Saint-Antonin, Stéphane Tchung-Ming and Simon Vinot. The usual disclaimer applies.

<sup>&</sup>lt;sup>†</sup>*Corresponding author*. IFP Energies nouvelles, 1-4 avenue de Bois-Préau, 92852 Rueil-Malmaison, France. benoit.cheze@ifpen.fr.

<sup>&</sup>lt;sup>‡</sup>EconomiX–CNRS, University of Paris Ouest, France.

<sup>&</sup>lt;sup>§</sup>Climate Economics Chair, Paris, France.

<sup>¶</sup>Ifsttar (LTE), Lyon, France; pascal.gastineau@ifsttar.fr.

<sup>&</sup>lt;sup>||</sup>Université Paris Dauphine (CGEMP/LEDa), France; julien.chevallier@dauphine.fr.

# 1 Introduction

Over the past thirty years, air traffic has been steadily increasing with growth rates dramatically superior to world's GDP growth rates. It is clearly established that the air transport sector has encountered during the second half of the  $20^{th}$  century a growth strictly superior to most sectors in the economy. This development may appear problematic during the  $21^{st}$  century (IPCC, 1999, 2007a, 2007b, 2007c; RCEP, 2002; ECI, 2006; IEA, 2009a, 2009b, 2009c). Be it because the fossil fuel resources are becoming scarce, or because of climate change, the evolution of air transport and its environmental consequences are now clearly taken into account by public policies. The classical example is the decision by the European Commission to include the aviation sector in the European Union Emissions Trading Scheme (EU ETS) during Phase III<sup>1</sup>.

Policy makers distinguish between two kinds of measures to limit the environmental impact of the growth of air transport. The first one consists in improving the energy efficiency of aircrafts (and thus to lower their carbon intensity) by promoting technological innovation. The second one consists in binding measures dealing with the demand for air transport. These two types of measures are not antinomic and may be pursued simultaneously. For instance, the European Commission actively promotes innovation efforts by air companies through R&D programs<sup>2</sup>, while also taking binding actions with the fight against climate change. However, these public policies do not follow the same logic. Indeed, the latter kind of measure deals directly with the causes of greenhouse gases emissions linked to the growth of the air transport sector, while the former kind of measure only attempts at diminishing its negative consequences.

There is a sharp debate today on the best measures to adopt in order to reduce significantly, and within a reasonable time horizon, greenhouse gases emissions from the air transport sector. Shall we solely aim at reducing the carbon intensity of the air transport sector through innovation? Or shall we admit that technological innovation only does not allow to achieve the objectives of stabilization and reduction of greenhouse gases that were set? In both cases,

<sup>&</sup>lt;sup>1</sup>As of 2012, air companies travelling in the EU must limit their  $CO_2$  emissions to 97% of the annual mean registered between 2004 and 2006. The airplanes taking off and landing in the EU 27, as well as coming from other countries, will fall under this regulation.

<sup>&</sup>lt;sup>2</sup>Two main projects are devoted currently to alternative fuels (other than Jet-Fuel) for aviation. The program Alfa-Bird, which was created in 2008 for a period of four years and coordinated by Airbus, aims at better identifying the needs in terms of research and investments. The Swafea program, led since January 2009 by the French Aerospace Lab, aims at elaborating a roadmap for deploying alternative fuels in the mid-term. Many industrial actors (Airbus, Avio, Dassault, EADS, Rolls-Royce, Safran, Sasoil, Shell, Snecma, etc.) are participating to these programs, along with public research centers.

shall we limit the development of the air transport sector?

This article aims at addressing these various questions by focusing on both estimating air traffic energy efficiency and energy gains. This task is performed based on the specific 'Traffic Efficiency method' developed by the UK DTI (Department of Trade and Industry) for the special IPCC report on air traffic (IPCC (1999)). The idea underlying this method may be summarized as follows. An increase by 5% per year of air traffic does not imply an increase by the same magnitude of Jet-Fuel demand, and thus corresponding  $CO_2$  emissions. Indeed, the growth of Jet-Fuel demand following the growth of air traffic is mitigated by energy efficiency gains<sup>3</sup>. Energy efficiency is obtained thanks to enhancements of (*i*) Air Traffic Management (ATM), (*ii*) existing aircrafts (changes in engines for example) and (*iii*) the production of more efficient aircrafts (which is linked to the rate of change of aircrafts)<sup>4</sup>.

Energy Efficiency is a measure for the technological performance of an individual aircraft or an aircraft fleet. Currently, no Energy Efficiency metric standard has been clearly established in the literature<sup>5</sup>. In this article, we have chosen to express Energy Efficiency in terms of mass of Jet-Fuel per Available Ton Kilometers  $(ATK)^6$ :

$$EE_{i,t} = \frac{Tjet_{i,t}}{ATK_{i,t}} \tag{1}$$

with  $EE_{i,t}$  the abbreviation for EE coefficient in zone i at time  $t^7$ . Thus defined, EE may

<sup>4</sup>See among others on this topic Greene (1992, 1996, 2004), IPCC (1999), Lee et al. (2001, 2004, 2009), Eyers et al. (2004), Lee (2010).

<sup>5</sup>According to Peeters et al. (2005), Lee et al. (2001) first introduced the term Energy Intensity (expressed in Mjoule/ASK) as a measure for the technological performance of an individual aircraft. Following Peeters et al. (2005), we prefer to use the term 'Energy Efficiency' rather than 'Energy Intensity'. Indeed, 'Energy Intensity' more refers to individual aircraft performances, whereas this study deals with estimating the actual efficiency of the collective fleet, *i.e.* on a global basis rather than at the aircraft level.

 $^6\mathrm{See}$  Owen (2008) for other EE metric definitions used in the literature.

<sup>7</sup>Revenue Ton Kilometer (RTK) is defined as one ton of load (passengers and/or cargo) carried for one kilometer. Available Ton Kilometers (ATK) is a measure of an airline's total capacity (both passenger and cargo). It is the capacity in tons multiplied by the number of kilometers flown. It would be natural to have RTK instead of ATK in this equation. However, before converting RTK into Jet-Fuel quantities, it is first necessary to convert RTK into ATK. The link between RTK and ATK is the Load Factor (LF), expressed in percentage. The latter may be defined as the percentage of an aircraft available ton effectively occupied during a flight. Thus for one flight,  $RTK = LF \times ATK$ .

 $<sup>^{3}</sup>$ For instance, over the last twenty years, the strong increase of air traffic has been accompanied by important progresses in the energy efficiency of aircrafts and aviation tasks (Greene (1992), Greene (2004)). Consequently, if Jet-Fuel demand has increased over the period, its growth rate has been largely lower than the air traffic one.

be interpreted as the quantity of Jet-Fuel (expressed in ton of Jet-Fuel) required to power the transportation of one ton over one kilometer  $(ATK)^8$ .

Thus, one of the major tasks of this article consists in examining the expected rates, expressed per year, of EE coefficients corresponding to the evolution of air traffic energy gains. According to previous literature, traffic efficiency improvements depend on: (i) load factors improvements (aircraft are using more of their capacity); (ii) energy efficiency improvements. In the former case, no technological progress is achieved: airlines diminish their Jet-Fuel consumption by filling more their aircrafts. In the latter case, there may be some opportunities for technological progress to happen. Energy efficiency improvements depend on a wide variety of factors, some of which are not linked to technological progress (such as Air Traffic Management), while others do. In the latter category, which is most likely predominant in the evolution of energy efficiency, the factors concern first the upgrade of existing aircrafts, and second changes in aircraft and airframe/engine design which are conditioned to the fleet renewal rate.

To estimate air traffic energy efficiency and energy gains, previous literature uses a specific methodology called 'bottom-up' in the remainder of the article. Nevertheless, this method has several limits that we will develop later. We propose to enhance this methodology with original and complementary ideas. The major contribution of this article consists in proposing a new methodology to obtain EE coefficients based on modeling at the macro-level. This new approach constitutes an important contribution, since it allows estimating the energy efficiency of aircraft fleets without assumptions on the actual composition of the fleet.

Our 'macro level' methodology allows obtaining 'aggregated' energy efficiency (EE) coefficients and their growth rates from 1983 to 2006. We notice that each of the regions considered have registered traffic efficiency improvements during the whole period at the aggregated (domestic+international) level. At the world level, energy efficiency improvements have been equal to 2.88% per year during the whole period. There are significant differences between regions. At the world level, domestic energy efficiency appears to be lower than the international one. This comment applies in all regions. This result confirms the intuition that domestic air travels are more energy intensive than international travels. One of the main reasons found in previous literature is that domestic flights are more energy intensive due to more frequent take-off and landing. These remarks lead to the following stylized fact: even if both international and domestic air travels have encountered energy efficiency improvements form 1983 to 2006, international air travels appear to be less energy intensive than domestic air travels.

<sup>&</sup>lt;sup>8</sup>Jet-Fuel consumption is obtained from the IEA, while ATK are given by the ICAO.

The remainder of the article is structured as follows. Section 2 summarizes previous 'bottom-up' methodologies. Section 3 introduces the new macro-level methodology. Section 4 contains the results from the new methodology. Section 5 compares the results of both approaches, by estimating the carbon intensity of several aircrafts and of the air transport sector as a whole. Section 6 concludes.

# 2 Methodologies used in previous literature: the 'Bottomup' approaches

Previous literature features two ways of modeling air transport mobility. First, modeling by routes (gravity models), and second modeling without routes (simple time-series analysis). In the former modeling, air traffic is estimated for various routes. At a more aggregated level, it allows to forecast traffic flows between two regions, for instance between Europe and Asia. On the contrary, the latter modeling does not allow to forecast traffic flows, but the expansion of various regions. In other words, the latter methodology provides spheres instead of routes.

To convert air transport traffic into Jet-Fuel demand, researchers generally use a 'bottomup' approach to (i) obtain EE coefficients, and (i) deduce an evolution rule for EE coefficients (see for instance Greene, 1992, 1996, 2004; IPCC, 1999; Eyers et al., 2004). This 'bottom-up' approach is mostly used for modeling by routes. In his econometric estimation of demand for air travel in the USA, Bhadra (2003) defines 'top-down' and 'bottom-up' approaches. When demand is determined econometrically by GDP, among other things, the estimated relationship is then allocated from the top down to the terminal areas, taking into consideration the historical shares of the airport, master plans, and expert opinion, to derive traffic forecasts. By contrast, when econometric relationships are estimated at a lower level (i.e., between origin and destination travel), they may be called a bottom-up approach. While traffic forecasts are primarily designed to serve as a terminal area planning tool, the latter approach focuses on market routes and flows (i.e., passengers and aircraft) within. Thus, 'bottom-up' approaches appear especially useful for network flow aspects. Several studies may be cited in this literature. Bhadra and Kee (2008) analyze the structure and dynamics of the origin and destination of core air travel market demand using 1995-2006 US quarterly time-series data. They show that passenger flows between origin and destination travel markets have exhibited strong growth in recent years. Macintosh and Wallace (2009) document international aviation emissions until 2025. They remark that the fuel efficiency gains associated with the latest generation of aircraft are unlikely to be sufficient to offset the increases in international demand, and conclude that the slow rate of turnover in the fleet will hinder progress on curbing emissions growth. Mazraati and Faquih (2008) model aviation fuel demand in the case of the USA and China. By estimating Jet-Fuel demand in these two extremes of a mature sector versus a quickly developing one, they confirm that mature sectors tend to be more sensitive to fluctuations in fuel prices and economic growth, as opposed to the quickly growing regions where the price effect is less pronounced<sup>9</sup>.

The so-called 'bottom-up' approach starts with the observation of aircrafts' energy efficiency (expressed in Mtoe/ASK, liter/ASK or Mjoule/ASK)<sup>10</sup>. Aircrafts' energy efficiencies are published by manufacturers. By replacing aircrafts' models by their vintage year, one can obtain (*i*) approximations of the values of Jet-Fuel consumption for a typical aircraft, and (*ii*) an idea of the evolution rule of EE coefficients overtime (Greene, 1992, 1996, 2004; IPCC, 1999; Eyers et al., 2004).

Such a representation is given in Figure 1. The first point represents the average Jet-Fuel consumption of the Comet 4 aircraft model issued in 1958. The last point represents the average Jet-Fuel consumption of the A350-900 aircraft model issued in 2011. In Figure 1, notice that due to technological innovations aircrafts' energy efficiency has been improving with a factor equal to 3.50 between 1958 and 2007.



Figure 1: Evolution of the average Jet-Fuel consumption by aircraft vintage expressed in Mjoule per ASK (1955-2010). Source: Authors, based on manufacturers' data.

Having detailed the 'bottom-up' methodology, one understands why it is usually used in

<sup>&</sup>lt;sup>9</sup>Besides, they show that the Chinese aviation sector and Jet-Fuel consumption will continue to outpace that of the United States, but growth in both regions will reach a steady state as the Chinese economy cools down and approaches maturity.

<sup>&</sup>lt;sup>10</sup>Available Seat Kilometer (ASK) measures an airline's passenger carrying capacity. It is: *seats available*  $\times$  *distance flown*.

the literature due to its intuitive appeal. However, this approach encounters several important empirical limits.

First, it relies on a few assumptions which may be seen as too restrictive. Indeed, once the 'bottom' step has been performed (as illustrated by Figure 1), some assumptions need to be made in order to obtain EE coefficients at the aggregated level. These assumptions include basically: (i) the composition of the aircrafts' fleet, and (ii) an evolution rule for this fleet concerning the renewal/upgrade policy of existing aircrafts. This underlying information about fleet characteristics and their evolution appears hard to investigate in practice, since researchers lack the access to detailed and reliable databases on this topic. The need for such data is all the more complicated that it is required by routes. Based on these restrictive assumptions, average aircrafts' Jet-Fuel consumption are used to obtain aggregated EE coefficients and their evolution rule.

Second, besides relying on restrictive assumptions, this approach is very time-consuming in terms of data management. Modeling by routes adds another layer of complexity, since this approach necessitates to obtain aggregated EE coefficients for each route.

Third, recall that there exist two main factors to increase traffic efficiency: load factors improvements on the one hand, and energy efficiency improvements on the other hand. The latter factor contains three possible sources of improvements: air traffic management (ATM), aircrafts' upgrades, and fleet renewal. Regarding energy efficiency improvements, the 'bottom-up' approach relies only on the last two sources. No improvements stemming from ATM can thus be accounted for when using this methodology.

Fourth, the last drawback concerns data availability. Recall that (i)  $EE_{i,t} = Tjet_{i,t}/ATK_{i,t}$ , and (ii) 'bottom-up' approaches are mostly used with modeling by routes. The International Civil Aviation Organization (ICAO) provides air traffic by routes only for international scheduled air traffic (not for domestic air traffic)<sup>11</sup>. The International Energy Agency (IEA) does not provide Jet-Fuel consumption by routes, but by countries. Whereas the 'bottom-up' approach leads to obtain Jet-Fuel consumption by routes, results cannot be confronted to actual data. Even if the 'bottom-up' approach is not used for modeling by route, it supposes to infer Jet-Fuel consumption data which is then adjusted to match historical data, as provided by the IEA.

Given these various limitations, an alternative methodology to compute directly aggregated EE coefficients is presented in the next section based on deductions from empirical

<sup>&</sup>lt;sup>11</sup>When forecasting Jet-Fuel demand at the worldwide level, this data limitation generates some incoherence in the methodology used: international air traffic may be modelled by route, while domestic air transport cannot. This limitation involves to use another type of dataset.

data.

## 3 Macro-level methodology proposal

This section proposes another approach to reconstruct EE coefficients values and their evolution rule. It departs from the previous methodology by (i) providing directly aggregated EE coefficients; and (ii) deducing them directly from empirical data.

As defined in eq(1):

$$EE_{i,t} = \frac{Tjet_{i,t}}{ATK_{i,t}}$$

where EE coefficients for the *i*-th region and date *t* correspond to the ratio of Jet-Fuel consumption  $(Tjet_{i,t})$  over air traffic  $(ATK_{i,t})$ .

The methodology proposed to obtain EE coefficients is to directly compare the Jet-Fuel consumption and the evolution of air traffic. As straightforward as it may look like, this methodology has not been implemented in such details before - to our best knowledge<sup>12</sup>.

Again, Jet-Fuel consumption is obtained from the IEA, while air traffic is given by the ICAO. More precisely the 'World Energy Statistics and Balances' database of the IEA provides Jet-Fuel consumption (expressed in ktoe) for the 1980–2006 period, while the 'Commercial Air Carriers - Traffic' database of the ICAO provides Air traffic (expressed in ATK) data during 1980–2007. Both databases provide these data by country. It is thus readily possible to re-aggregate these time-series by region.

Fo our analysis, we choose the following geographical decomposition: Central and North America, Latin America, Europe, Russia and CIS (Commonwealth of Independent States), Africa, the Middle East, China<sup>13</sup>, Asian countries and Oceania. Worldwide basis corresponds to the computation of these eight distinct regions.

This macro-level methodology allows then to obtain the 'aggregated' EE coefficients – as opposed to 'bottom-up' EE coefficients – and their growth rates from 1980 to 2006. This idea is summarized for a typical region in Figure 2. On the left panel, the solid black line represents air traffic (expressed in ATK) and the dotted black line represents Jet-Fuel consumption (expressed in ktoe) for a given region. As defined in eq(1), EE coefficients for each year may be obtained by dividing ktoe/ATK (right panel).

Thus defined, EE corresponds to the quantity of Jet-Fuel required to power the transportation of one ton over one kilometer. For a given region  $EE_{t+1} < EE_t$  means that quantities of

 $<sup>^{12}</sup>$ Peeters et al. (2005) and Owen (2008) already had the same intuition, but they did not apply the methodology at such a detailed level compared to what we do here.

<sup>&</sup>lt;sup>13</sup>We chose to focus on that specific region due to its solid economic growth.



Left panel: Jet-Fuel consumption (expressed in Mtoe) and air traffic (expressed in ATK). Right panel: EE coefficients computed as the ratio of the former over the latter.

Figure 2: Illustration of the Macro-level methodology used to compute 'aggregated' EE coefficients and their yearly growth rates.

Jet-Fuel required to power the transportation of one ton over one kilometer have decreased. Thus, a negative growth rate of EE coefficients, as it is expected, indicates the realization of energy efficiency improvements in air traffic for the region under consideration. As it may be deduced from the illustrative Figure 2, EE coefficients negative growth rates arise when, in a given year, Jet-Fuel consumption growth rates are slower than air traffic ones.

By following this methodology, first for each zone the value of the EE coefficients until 2006 is obtained. Second, an evolution rule for these EE coefficients in the future may be derived for each zone by observing the evolution of their growth rates between 1980 and 2006. Actually, both datasets are available at an even more disaggregated level for each zone, *i.e.* domestic *vs.* international. Following the same methodology for each region, it becomes thus possible to obtain not only the 'aggregated' EE coefficients, but also EE coefficients corresponding to both international and domestic air travels.

This methodology allows to investigate three issues. First, by comparing the evolution of EE coefficients overtime, one may observe the occurrence (or not) of energy efficiency improvements over the last 30 years. Second, by comparing the values found for aggregated EE coefficients, one may deduce which zone is more energy efficient compared to others. Third, by comparing 'domestic' and 'international' EE coefficients within each zone, one may observe if domestic air travel is effectively less efficient than international air travel<sup>14</sup>. These questions are investigated in-depth in Section 4.

The new methodology proposed seems promising. However, it is also characterized by

<sup>&</sup>lt;sup>14</sup>As highlighted in the literature (Gately, 1988; Vedantham and Oppenheimer, 1998), domestic air traffic is supposed to be more energy intensive than international air traffic due to more frequent take-off and landing of aircrafts, the most energy-intensive components of a flight.

some limitations.

First, EE coefficients obtained cannot be used in a modeling by routes. This restriction supposes a modeling without routes, as done in this article. This corresponds to an output loss compared to the 'bottom-up' approach, which does not prevent from using either of the two modeling types of air transport mobility.

Second, even if all potential sources of energy efficiency improvements are covered by the macro-level methodology, it is not possible to disentangle the effects from which improvements in energy efficiency are obtained. Recall that it could come from ATM, aircrafts' upgrades, aircraft and airframe/engine design (which is linked to fleet renewal rates). However, this drawback is relatively less important than the corresponding limitations of the 'bottom-up' approach, which cannot account for the ATM source of possible energy efficiency improvements.

Overall, each methodology ('bottom-up' vs. 'macro-level') involves numerous assumptions. Results of the new methodology are given in the next section.

## 4 Results of the Macro-level methodology

Databases are first re-aggregated by region. Then, EE coefficients are computed for each region. Countries do not necessarily start declaring their data simultaneously. For instance, China has started to declare its air traffic data to the ICAO since 1993. As a consequence, exogenous shocks in the evolution of EE coefficients values may be wrongly interpreted, as they only reflect the entrance of a new data source (*e.g.* a country starts declaring either its Jet-Fuel consumption or its air traffic data). Thus, to smooth these potential biases in the data, EE coefficients are presented in mean values during two sub-periods: 1983-1996 and 1996-2006, besides the whole period.

The USSR started to declare its air traffic data in 1983 only. Besides the USSR, other countries did not declare either air traffic data or Jet-Fuel consumption during the first years of the 1980s. Thus, it has been chosen to start the first sub-period in 1983, in particular to allow comparisons of the Russia and CIS region with other regions.

EE mean values during the first sub-period are not provided for two regions: China, and Russia and CIS. Again, China starts declaring its air traffic data in 1993. Russia and CIS presents some inconsistencies in the data during 1991-1992, since this region had to be re-aggregated.

This section presents results from the macro-level methodology. A three-step analysis is conducted here.

First, EE coefficients values for each zone and the world and their respective growth rates are presented and analyzed. By comparing the evolution of EE coefficients overtime, one may address both research questions, *i.e.* what is the value of the EE coefficients for each zone, and what is their respective evolution rule. These coefficients are given for international and domestic travels, and at the aggregated (domestic + international) level.

Second, EE coefficients values are compared in order to assess which region is more energy efficient compared to the world's average.

Third, within each zone, domestic EE coefficients are compared with international EE coefficients. This is done to test if domestic air travel is less energy efficient than international air travel, as underlined in the literature.

It is worthy to remark that, to our best knowledge, this article provides for the first time EE coefficients at such a detailed level: (i) by region; and (ii) by type of travel (domestic vs. international).

# 4.1 How do EE coefficients evolve overtime? An analysis for each zone and worldwide

EE coefficients mean values, their yearly mean growth rates for sub-periods and the whole period, and the rate of change during the whole period are provided in Table 1. These coefficients are presented for domestic travel, international travel, and aggregated (domestic+international) travel, and for each region and the world. Comments are not provided for the mean value of each zone, as the actual figures obtained are not meaningful. However, the comparison of these coefficients between and within regions yields significant economic insights. These comments are presented in the two next subsections (respectively in Tables 2 and 3)<sup>15</sup>.

In what follows, only yearly mean growth rates are commented upon. EE coefficients indicate the quantities of Jet-Fuel required to power the transportation of one ton over one kilometer (recall eq(1)). Hence computed, a decrease in EE coefficients indicates that less Jet-Fuel is needed to power the same unit of air transport. Thus, negative growth rates of EE coefficients shall be interpreted as energy efficiency improvements.

All regions have registered energy efficiency improvements during the whole period at the aggregated (domestic+international) level. As shown in Table 1, all yearly mean growth rates

<sup>&</sup>lt;sup>15</sup>As explained in the introduction, some authors rather express energy efficiency coefficients as the ratio of Jet-Fuel consumption over air traffic  $(EE'_{i,t} = \frac{T_jet_{i,t}}{ATK_{i,t}})$ . In this case, one generally prefers to use the term 'Traffic Efficiency' (see Owen (2008) for more details). Traffic efficiency is then the reciprocal of fuel efficiency.

		Mean values			Yearly average growth rates (EE gains)			Rate of change
		Sub-periods Whole period		${f Sub-periods}$		Whole period		
		1983-1996	1996-2006	1983-2006	1983-1996	1996-2006	1983-2006	1983-2006
Central and	Aggregated	3.93E-07	2.90E-07	3.49E-07	-1.78%	-3.18%	-2.39%	-42.65%
North	Domestic	4.58E-07	3.62E-07	4.16E-07	-1.71%	-1.86%	-1.78%	-33.80%
America	International	2.60E-07	1.80E-07	2.25E-07	-1.04%	-5.27%	-2.91%	-49.25%
Europe	Aggregated	3.52E-07	2.71E-07	3.18E-07	-2.97%	-1.20%	-2.20%	-40.10%
	Domestic	8.75E-07	7.31E-07	8.17E-07	-3.99%	1.40%	-1.68%	-32.35%
	International	3.02E-07	2.35E-07	2.74E-07	-2.58%	-1.25%	-2.00%	-37.22%
Latin	Aggregated	4.22E-07	4.35E-07	4.31E-07	-3.73%	1.18%	-1.63%	-31.42%
America	Domestic	7.21E-07	6.24E-07	6.81E-07	-4.05%	-3.81%	-3.95%	-60.41%
	International	2.85 E-07	3.31E-07	3.08E-07	-3.46%	5.03%	0.14%	3.34%
Russia and	Aggregated	n.a.	1.00E-06	n.a.	n.a.	-5.79%	n.a.	-44.92% *
CIS	Domestic	n.a.	2.09E-06	n.a.	n.a.	-5.37%	n.a.	-42.39% *
	International	n.a.	6.89E-07	n.a.	n.a.	-5.86%	n.a.	-45.33% *
Africa	Aggregated	7.81E-07	9.18E-07	8.30E-07	4.45%	-7.22%	-0.80%	-16.79%
	Domestic	1.80E-06	3.94E-06	2.69E-06	12.51%	-7.14%	3.50%	120.60%
	International	6.60 E- 07	6.78E-07	6.62 E- 07	2.65%	-7.63%	-1.95%	-36.43%
The Middle	Aggregated	6.75E-07	5.07E-07	6.02E-07	0.02%	-8.68%	-3.86%	-59.56%
East	Domestic	5.53E-07	1.00E-06	7.36E-07	8.40%	-11.23%	-0.62%	-13.29%
	International	7.08E-07	4.87E-07	6.14E-07	-0.79%	-8.46%	-4.20%	-62.75%
Asian	Aggregated	3.17E-07	2.44E-07	2.85E-07	-2.88%	-1.54%	-2.30%	-41.46%
countries and	Domestic	5.87E-07	4.03E-07	5.08E-07	-6.31%	-2.80%	-4.80%	-67.73%
Oceania	International	2.69E-07	2.10E-07	2.44E-07	-2.35%	-0.79%	-1.67%	-32.18%
China	Aggregated	n.a.	2.22E-07	n.a.	n.a.	-1.65%	n.a.	-15.37% *
	Domestic	n.a.	3.53E-07	n.a.	n.a.	-2.37%	n.a.	-21.32% *
	International	n.a.	1.56E-07	n.a.	n.a.	-2.45%	n.a.	-21.94% *
World	Aggregated	4.17E-07	2.98E-07	3.66E-07	-3.09%	-2.61%	-2.88%	-48.95%
	Domestic	4.52E-07	4.17E-07	4.36E-07	-0.20%	-1.95%	-0.96%	-19.94%
	International	3.96E-07	2.35E-07	3.28E-07	-5.23%	-2.56%	-4.08%	-61.62%
	-mer national	5.001 01	2.001 01	0.201 01	0.2070	2.0070	1.0070	1 01.0270

Note: \* means that rates of change are not computed for the whole period, but for the second sub-period.

Table 1: EE coefficients (ktoe/ATK) for each zone and worldwide. Means values and growth rates during 1983-2006. Source: Authors, from ICAO and IEA data.



Figure 3: Jet Fuel consumption (Mtoe), ATK (billions) and EE coefficient at the world level. Source: Authors, from ICAO and IEA data.

are negative, ranging from -0.80% (Africa) to -3.86% (the Middle East)<sup>16</sup>. At the world level, energy efficiency improvements have been equal to 2.88%/yr during the whole period. As illustrated by Figure 3, still at the world level, energy efficiency improvements have been more important during 1983-1996 (3.09%/yr) than during 1996-2006 (2.61%/yr).

The macro-level methodology proposed here leads us to recover, and quantify, previous results highlighted in the literature. Energy efficiency improvements have been effectively accomplished in the air transport sector. According to our methodology, these energy efficiency improvements have been rather important during the last 30 years (about 3%/yr at the world level).

These results depart however from previous literature. First, energy efficiency improvement values drawn from the macro-level approach are relatively higher than those obtained with the 'Bottom-up' method. Indeed, the most often cited energy efficiency gains estimates are generally comprised between 1.5%/yr (Lee et al., 2004) and 2.2%/yr (Airbus,  $2007)^{17}$ . Second, applied to the eight different regions, the macro-level methodology indicates that energy efficiency improvements have been very heterogeneous between regions during the last 30 years.

The next two sections present the comparison between and within regions of these EE coefficients values.

<sup>&</sup>lt;sup>16</sup>Note the presence of two outliers at the domestic vs. international level: Africa registers a yearly mean growth rate of +3.50% at the domestic level during the whole period (this region records however negative yearly mean growth rates during the second sub-period), and Latin America registers a positive growth rate of +0.14% at the international level during the whole period.

<sup>&</sup>lt;sup>17</sup>See Eyers et al. (2004) and Mayor and Tol (2010) for a more comprehensive literature review.

#### 4.2 Which region is more energy efficient?

To compare EE coefficients between regions, three kinds of ratios between EE coefficients are computed. Results are presented in Table 2.

In Table 2, aggregated (domestic + international), domestic and international EE coefficients mean values of each region are compared to the world mean values for the whole and the corresponding sub-periods. The ratios presented in the first (respectively second and third) line of the *i*-th region correspond to, for the period under consideration, the aggregated (respectively domestic and international) EE coefficient mean value of the *i*-th region over the aggregated (respectively domestic and international) EE coefficient mean value of the *w*orld. In other words, these ratios are computed as follows:

$$\frac{EE_{i,t,k}}{EE_{w,t,k}}\tag{2}$$

where  $EE_{i,t,k}$  represents the EE coefficient mean value of region *i*, at time  $t = \{1983-1996;1996-2006;1983-2006\}$ , and for the type of travel  $k = \{aggregated; domestic; international\}$ .  $EE_{w,t,k}$  represents the EE coefficient mean value of the world, at time  $t = \{1983-1996;1996-2006;1983-2006\}$ , and for the type of travel  $k = \{aggregated; domestic; international\}$ .

For instance, the value in the first line of the first column (0.95) represents the relative energy efficiency mean value of the Central and North American region during 1983-1996, when compared to the world's energy efficiency. It corresponds to the ratio of 3.93E - 0.7/4.17E - 0.7, where 3.93E-0.7 is equal to the Central and North American region EE coefficient value during 1983-1996, and 4.17E - 0.7 is equal to the World's EE coefficient value during 1983-1996 (Table 1).

A ratio superior to one means that one needs more quantity of Jet-Fuel to transport one ton kilometer in a given region compared to the world's average. Thus constructed, a ratio >(<) 1 means that the region's energy efficiency is inferior (superior) to the world's energy efficiency.

During the whole period<sup>18</sup>, aggregated (domestic + international) EE ratios are less than one for four regions (Central and North America, Europe, China, Asia and Oceania), and greater than one for the four others (Latin America, Africa, Russia and CIS, the Middle East) (Table 2). This result means that, for aggregated (domestic + international) travel, the former regions are in average more energy efficient during the whole period than the world's

<sup>&</sup>lt;sup>18</sup>Comments apply only for the second sub-period for Russia and CIS, and China. See above in Section 4 for more details.

		Mean values			Yearly average growth rates			Rate of change
		Sub-p	eriods	Whole period	Sub-p	eriods	Whole period	
		1983-1996	1996-2006	1983-2006	1983-1996	1996-2006	1983-2006	1983-2006
Central and	Zone's aggregated EE / World's aggregated EE	0.95	0.97	0.96	1.36%	-0.59%	0.51%	12.34%
North	Zone's domestic EE / World's domestic EE	1.01	0.87	0.95	-1.52%	0.09%	-0.82%	-17.31%
America	Zone's international EE / World's international EE	0.69	0.76	0.71	4.41%	-2.78%	1.22%	32.24%
Europe	Zone's aggregated EE / World's aggregated EE	0.85	0.91	0.88	0.13%	1.44%	0.70%	17.33%
	Zone's domestic EE / World's domestic EE	1.94	1.76	1.87	-3.80%	3.41%	-0.73%	-15.50%
	Zone's international EE / World's international EE	0.79	1.00	0.88	2.79%	1.35%	2.16%	63.58%
Latin	Zone's aggregated EE $/$ World's aggregated EE	1.00	1.49	1.22	-0.66%	3.88%	1.29%	34.33%
America	Zone's domestic EE / World's domestic EE	1.59	1.50	1.56	-3.86%	-1.90%	-3.02%	-50.55%
	Zone's international EE / World's international EE	0.74	1.45	1.05	1.86%	7.79%	4.40%	169.25%
Russia and	Zone's aggregated EE / World's aggregated EE	n.a.	3.34	n.a.	n.a.	-3.27%	n.a.	-28.26% *
CIS	Zone's domestic EE / World's domestic EE	n.a.	4.95	n.a.	n.a.	-3.49%	n.a.	-29.87% *
	Zone's international EE / World's international EE	n.a.	2.91	n.a.	n.a.	-3.38%	n.a.	-29.12% *
Africa	Zone's aggregated EE / World's aggregated EE	1.95	3.03	2.39	7.78%	-4.74%	2.15%	62.99%
	Zone's domestic EE / World's domestic EE	4.00	9.27	6.22	12.73%	-5.30%	4.51%	175.54%
	Zone's international EE $/$ World's international EE	1.80	2.83	2.21	8.31%	-5.20%	2.22%	65.63%
The Middle	Zone's aggregated EE / World's aggregated EE	1.66	1.67	1.66	3.21%	-6.24%	-1.01%	-20.78%
East	Zone's domestic EE / World's domestic EE	1.23	2.37	1.71	8.61%	-9.46%	0.35%	8.31%
	Zone's international EE $/$ World's international EE	1.90	2.04	1.95	4.68%	-6.06%	-0.13%	-2.95%
Asian	Zone's aggregated EE / World's aggregated EE	0.76	0.82	0.79	0.21%	1.10%	0.60%	14.66%
countries and	Zone's domestic EE / World's domestic EE	1.29	0.96	1.15	-6.12%	-0.87%	-3.87%	-59.70%
Oceania	Zone's international EE / World's international EE	0.70	0.90	0.79	3.04%	1.82%	2.51%	76.71%
China	Zone's aggregated EE / World's aggregated EE	n.a.	0.75	n.a.	n.a.	0.98%	n.a.	10.22% *
	Zone's domestic EE / World's domestic EE	n.a.	0.81	n.a.	n.a.	-0.43%	n.a.	-4.22% *
	Zone's international EE $/$ World's international EE	n.a.	0.67	n.a.	n.a.	0.12%	n.a.	1.19% *

Note: a ratio >(<) 1 means that the region's energy efficiency is inferior (superior) to the world's energy efficiency. These ratios are provided for the aggregated (domestic+international), domestic, and international travels.

\* means that rates of change are not computed for the whole period, but for the second sub-period.

Table 2: Comparison of EE coefficients (ktoe/ATK) between zones using world's EE coefficients as benchmark (1983-2006). Source: Authors, from ICAO and IEA data.

15

benchmark. On the contrary, the four latter regions are less energy efficient than the world's average during 1983-2006. According to previous literature (Greene, 1992, 1996, 2004; IPCC, 1999; Eyers et al., 2004), these results appear quite intuitive except for the Middle East region. Indeed, the Middle East seems to be 1.66 more energy-intensive than the world's benchmark (Table 2). This particular case is further investigated below by a visual inspection of the data. Comments are not further developed at the domestic vs. international level, since they follow the same trends as observed at the aggregated (domestic + international) level.

Figure 4(a) provides a visual representation of the evolution of EE coefficients. It compares each region's aggregated EE coefficients against the world's benchmark. To understand the evolution of EE coefficients, one needs to know the evolution of these two time-series (Figure 4(b-c)). By looking at Figure 4, one may observe the results commented in Table 2. EE coefficients of Central and North America, Europe, Asia and Oceania and China (Figure 4(a), left panel) are globally below the EE world's benchmark. One retrieves indeed the result that these regions are the least energy-intensive in the world. Similarly, the same patterns as in Table 2 are observable for the four more energy-intensive regions. Figure 4 provides an additional information compared to Table 2: all EE trends are decreasing globally. These trends confirm that each region has achieved energy efficiency improvements.

Figure 4 allows to understand the evolution of EE coefficients by representing the evolution of its constituent aggregates: Jet-Fuel consumption (expressed in Mtoe, Figure 4(b)) and air traffic (expressed in ATK, Figure 4(c)). This representation is convenient, since it may explain the *a priori* counter-intuitive results observed in the Middle East. Indeed, Table 2 indicated that this region is less energy efficient than the world's benchmark. It is common knowledge that airline companies in the Middle East are currently purchasing new aircrafts. Thus, they have a higher fleet renewal rate than other airlines. One may deduce that in this region the performance in terms of energy efficiency should be relatively better than the world's benchmark. By looking at the right panel of Figure 4, EE coefficients are effectively always above the world's benchmark during the period, but they have dramatically decreased since 2001 to be below this benchmark in 2006. When looking at the right panel of Figure 4, a strong increase of the traffic registered in this region may be noted since 2001. However, one cannot notice an equivalent increase in the consumption of Jet-Fuel during the same period, which means that energy efficiency improvements must have occurred through the use of newer aircrafts.

The next section compares international and domestic EE coefficients.



 — China
 — The Middle East

 — Europe
 — Africa

 — Asian countries and Oceania
 — Latin America

 — North and Central America
 — Russia and CIS

Note: China starts declaring some of its air traffic data in 1993. Russia and CIS presents some inconsistency in the data until 1991. Thus, some statistics must be interpreted with great care.

Figure 4: Aggregated (domestic+international) EE coefficients by region against the world; evolution of Jet-Fuel consumption (expressed in Mtoe) and air traffic (expressed in ATK (billions)) by region. Source: Authors, from ICAO and IEA data.

# 4.3 Are domestic air travels less energy efficient than international ones?

To reply to this question, one proposes to compare EE coefficients within regions. To do so, three kinds of ratios between EE coefficients are computed. Results are presented in Table 3.

In Table 3, within each zone, domestic and international EE coefficients mean values are compared to respectively aggregated (domestic + international) and international mean values for the whole and the corresponding sub-periods. Ratios presented in the first (respectively second and third) line of the *i*-th region correspond to, for the period under consideration, the domestic (respectively international and domestic) EE coefficient mean value of the *i*-th region over the aggregated (respectively aggregated and international) EE coefficient mean value of the same region. In other words, these ratios are computed as follows:

$$First \ Ratio = \frac{EE_{i,t,dom}}{EE_{i,t,agg}} \quad ; \quad Second \ Ratio = \frac{EE_{i,t,int}}{EE_{i,t,agg}} \quad ; \quad Third \ Ratio = \frac{EE_{i,t,dom}}{EE_{i,t,int}} \quad (3)$$

where:

 $EE_{i,t,dom}$  represents the EE coefficient mean value of region *i*, at time  $t = \{1983-1996; 1996-2006; 1983-2006\}$  for domestic air travel;

 $EE_{i,t,agg}$  represents the EE coefficient mean value of region *i*, at time  $t = \{1983-1996; 1996-2006; 1983-2006\}$  for aggregated (domestic + international) air travel;

 $EE_{i,t,int}$  represents the EE coefficient mean value of region *i*, at time  $t = \{1983-1996; 1996-2006; 1983-2006\}$  for international air travel.

Thus constructed, a ratio >(<) 1 means that the energy efficiency of the kind of travel in numerator is inferior (superior) to the kind of travel in denominator. These ratios aim at comparing, within each region, (i) the domestic vs. aggregated (domestic+international) EE coefficients mean values, (ii) the international vs. aggregated (domestic+international) EE coefficients mean values, and (iii) the domestic vs. international EE coefficients mean values.

At the world level, domestic energy efficiency appears to be lower than at the international level. There is a ratio of 1.33 to one between world's international and domestic energy efficiencies for the whole period (Table 3). This comment applies in all regions: domestic energy efficiency appears to be inferior to international energy efficiency whatever the region considered (third line for each zone). This result confirms the intuition that domestic air travels are more energy intensive than international air traffic. One of the main reasons found in previous literature is that domestic flights are more energy intensive due to more frequent take-off and landing.

		Mean values			Yearly average growth rates			Rate of change
		Sub-periods		Whole period	Sub-p	eriods	Whole period	
		1983-1996	1996-2006	1983-2006	1983-1996	1996-2006	1983-2006	1983-2006
Central and	Zone's domestic EE / Zone's aggregated EE	1.16	1.25	1.20	0.06%	1.36%	0.63%	15.44%
North	Zone's international EE $/$ Zone's aggregated EE	0.66	0.62	0.64	0.74%	-2.16%	-0.53%	-11.50%
America	Zone's domestic EE / Zone's international EE	1.77	2.06	1.85	-0.68%	3.60%	1.16%	30.44%
Europe	Zone's domestic EE / Zone's aggregated EE	2.46	2.71	2.57	-1.05%	2.63%	0.53%	12.94%
	Zone's international EE / Zone's aggregated EE	0.86	0.87	0.86	0.40%	-0.05%	0.20%	4.81%
	Zone's domestic EE / Zone's international EE	2.87	3.13	2.99	-1.45%	2.68%	0.33%	7.76%
Latin	Zone's domestic EE / Zone's aggregated EE	1.69	1.44	1.57	-0.34%	-4.93%	-2.36%	-42.27%
America	Zone's international EE / Zone's aggregated EE	0.68	0.75	0.72	0.28%	3.81%	1.80%	50.69%
	Zone's domestic EE / Zone's international EE	2.53	1.89	2.21	-0.61%	-8.42%	-4.09%	-61.69%
Bussia and	Zone's domestic EE / Zone's aggregated EE	na	2.04	na	na	0.45%	na	4 59% *
CIS	Zone's international EE / Zone's aggregated EE	na	0.69	na	na	-0.07%	na	-0.75% *
015	Zone's domestic EE / Zone's international EE	n a	2.99	n a	n a	0.53%	n a	5.38% *
		11.4.	2.00	11.6.	in.a.	0.0070	11.4.	0.0070
Africa	Zone's domestic EE / Zone's aggregated EE	2.30	4.29	3.24	7.71%	0.09%	4.33%	165.11%
	Zone's international EE $/$ Zone's aggregated EE	0.86	0.74	0.81	-1.72%	-0.43%	-1.16%	-23.60%
	Zone's domestic EE / Zone's international EE	2.72	5.81	4.06	9.60%	0.53%	5.56%	247.03%
The Middle	Zone's domestic EE / Zone's aggregated EE	0.82	1.93	1.28	8.37%	-2.79%	3.37%	114.41%
East	Zone's international EE / Zone's aggregated EE	1.05	0.96	1.01	-0.81%	0.24%	-0.36%	-7.91%
	Zone's domestic EE / Zone's international EE	0.80	2.02	1.21	9.26%	-3.02%	3.74%	132.81%
Asian	Zone's domestic EE / Zone's aggregated EE	1.81	1.65	1.74	-3.52%	-1.28%	-2.56%	-44.88%
countries and	Zone's international EE / Zone's aggregated EE	0.85	0.87	0.86	0.55%	0.76%	0.64%	15.86%
Oceania	Zone's domestic EE / Zone's international EE	2.15	1.91	2.05	-4.05%	-2.03%	-3.18%	-52.43%
			1 50			0 500		
China	Zone's domestic EE / Zone's aggregated EE	n.a.	1.58	n.a.	n.a.	-0.73%	n.a.	-7.03% *
	Zone's international EE / Zone's aggregated EE	n.a.	0.70	n.a.	n.a.	-0.81%	n.a.	-7.77% *
	Zone's domestic EE / Zone's international EE	n.a.	2.27	n.a.	n.a.	0.08%	n.a.	0.80% *
World	Zone's domestic EE / Zone's aggregated EE	1.10	1.41	1.23	2.99%	0.68%	1.98%	56.83%
	Zone's international EE $/$ Zone's aggregated EE	0.94	0.79	0.88	-2.21%	0.05%	-1.23%	-24.82%
	Zone's domestic EE / Zone's international EE	1.14	1.78	1.33	5.31%	0.63%	3.25%	108.60%

Note: a ratio >(<) 1 means that the energy efficiency of the kind of travel in numerator is inferior (superior) to the kind of travel in denominator. These ratios aim at comparing. within each region, (i) the domestic vs. aggregated (domestic+international) EE coefficients mean values, (ii) the international vs. aggregated (domestic+international) EE coefficients mean values, and (iii) the domestic vs. international EE coefficients mean values.

\* means that rates of change are not computed for the whole period, but for the second sub-period.

Table 3: Comparison of domestic and international EE coefficients (ktoe/ATK) within each zone (1983-2006). Source: Authors, from ICAO data.



Figure 5: Comparison of the evolution of (i) aggregated (domestic + international), (ii) domestic and (iii) international EE coefficients at the world level from 1990 to 2006. Source: Authors, from ICAO and IEA data.

Figure 5 clearly illustrates this stylized fact. At the world level, international air travels (black dashed line) are more energy efficient than domestic air travels (gray dashed line), over the last twenty years. Indeed, the curve of domestic EE coefficients is above the curve for international EE coefficients<sup>19</sup>. Thus, this figure illustrates previous results presented in Table 3. Moreover, the decreasing trend of the three curves illustrates the results presented in Table 1: both international and domestic air travels – and as a consequence aggregated (domestic + international) air travel too – have encountered energy efficiency improvements during 1983-2006 at the world level. The same kind of figures may be obtained at the regional level. They are not provided here as they would exhibit exactly the same kind of pattern and stylized fact<sup>20</sup>.

The two precedent remarks lead to the following stylized fact: even if both international and domestic air travels have encountered energy efficiency improvements from 1983 to 2006, international air travels appear to be less energy intensive than domestic air travels. The macro-level approach proposed in this article conducts to the same conclusions as in previous literature, but obtained with 'bottom-up' approaches. Applied to air traffic at the world level, the macro-level approach allows to quantify this stylized fact: air traffic efficiency gains have been equal to +4.08%/yr and +1.00%/yr during the whole period, respectively for

 $<sup>^{19}</sup>$ As a consequence the curve of aggregated (domestic + international) EE coefficients (solid black line) is between the two other ones.

<sup>&</sup>lt;sup>20</sup>These figures may be obtained upon request to the authors.

international and domestic air travels (Table 1).

This macro-level methodology has been initially developed to deduce air transport energy efficiency improvements *scenarii* in order to forecast Jet-Fuel consumptions and  $CO_2$  emissions from aviation. As already explained, the first interest of the macro-level methodology is to obtain precisely quantified results regarding air transport energy efficiency improvements both at the world level and at a more disaggregated level (the eight regions). Beyond this primary important output, the interest of this macro-level methodology is reinforced by the opportunity to make use of these results in many different ways. The next section illustrates this argument.

# 5 Comparison of estimates provided by 'bottom-up' and 'macro' methodologies

These two methodologies are designed to investigate whether air transport has experienced (or not) an improvement in energy efficiency during last decades. They aim (i) at estimating the absolute levels of energy consumption of air transport, and (ii) at quantifying the potential improvement. The next two subsections compare the estimates provided by these two methodologies. These comparisons allow us to highlight the complementarities and limitations of both methods.

### 5.1 Evolution of energy efficiency of air transport

Regarding the improvement of energy efficiency in air transport, our estimates are higher than those obtained using the 'Bottom-up' methodology. Nevertheless we find the same trend, i.e. energy efficiency in air transport has improved but at a slower pace since the 1980s.

#### 5.1.1 Energy efficiency of air transport has improved in the last decades ...

The studies using the 'Bottom-up' methodology highlight an improvement in energy efficiency of air transport in recent years (Figure 1). Most studies refer to the previous exhaustive research of Greene (1992). Based on this study, the IPCC special report on aviation (IPCC, 1999) estimated that energy efficiency of air transport has improved by about 70% between 1960 and 2000 (i.e. 3%/yr). Nevertheless, some authors (Peeters et al., 2005; Owen, 2008) argued that a methodological bias may have led to an overestimation of improvement of energy efficiency by Greene (1992). Most values are between 1.5%/yr (Lee et al., 2004) and 2.2%/yr (Airbus, 2007). These values are lower than ours (2.6%/yr over the last ten years, see Table 1). An explanation for these differences may be found at the end of the next subsection.

#### 5.1.2 ... but at decreasing pace

Most studies emphasize the fact that energy efficiency of air transport has improved significantly, but that this pace has slowed. Figure 1 shows the average energy consumption (expressed in Mega-Joules (MJ) per ASK) for different types of aircrafts. It reveals the discontinuity in the improvement of aircrafts' energy efficiency. The sharper improvement between 1950-1980 can be explained by the development of turbojet and turbofan engines (Lee et al., 2001, 2004; Owen, 2008). The pace has been slower during the last 30 years (Figure 1 and/or Table 4).

At the world level, as illustrated by Table 1, the energy efficiency of air transport has been improving by 2.9%/yr between 1983 and 2006, but it was stronger during the period 1983-1996 (3.1%/yr) than between the period 1996-2006 (2.6%/yr). At the regional level, we find the same evidence (Table 1). Nevertheless, we cannot compare our regional estimates with the 'Bottom up' methodology, as this methodology only provides estimates of energy efficiency for a representative world fleet.

We need to remain cautious when comparing the estimates obtained between both methods, as they do not necessarily measure the same phenomenon. Contrary to our methodology, the 'Bottom-Up' approaches only estimates the aircraft's energy efficiency. This methodology does not take into account the effect of ATM in the calculation of energy savings. Hence, the deviation estimates observed between the two methodologies may be partly explained by the non-inclusion of the ATM in the 'Bottom-up' methodology. Thus, these differences underline the need to take into account ATM improvements when estimating energy efficiency gains, and their evolution in air transport.

Another limitation of the 'Bottom-up' methodology lies in its implementation. Assumptions about the composition of the fleet necessarily rely on arbitrary choices. The energy efficiency of the fleet is therefore dependent on these choices. Instead, the 'macro-level' methodology computes the energy efficiency of air transport from data about (i) actual Jet-Fuel consumption, and (ii) air traffic actually observed. By doing so, we can estimate the absolute level of energy efficiency of the world fleet without having to detail the composition of the fleet.

Thus, the two approaches appear rather complementary. The next subsection aims at comparing the absolute levels of energy efficiency that these two methodologies provide.

#### 5.2 Carbon intensity of air transport

Fuel consumption and emissions are two major issues when designing an aircraft. Nevertheless, energy efficiency is not the only parameter that matters as shown by its slight improvement since the 1980's. As underlined by Peeters et al. (2005), airlines buying new aircrafts consider several other parameters such as cruising speed (Frederick, 1961), comfort, or the maximum capacity of the unit. The aircraft A-380 illustrates this fact: its energy consumption is not always better than some models developed twenty years earlier (see Figure 1 and Table 4).

We now compare the average carbon intensity of each aircraft with the average carbon intensity of the world civil fleet. These figures are shown in Tables 4 and 5. Table 4 is obtained by using the 'Bottom-up' methodology, while Table 5 is obtained by using the 'macro-level' methodology<sup>21</sup>.

As illustrated by Tables 4 and 5, the two methodologies provide complementary information, and their joint analysis may improve the understanding of the energy efficiency in air transport. The 'Bottom-up' methodology provides the carbon intensity for each aircraft, while the 'macro-level' approach provides carbon intensity for flying fleets. There are genuine differences between these estimates. For example, the A-380 is 60% less carbon-intensive than the world fleet (78 g-CO<sub>2</sub>/ASK, see Table 4 *vs.* 126 g-CO<sub>2</sub>/ASK, see Table 5)<sup>22</sup>.

Tables 4 and 5 show Jet-Fuel consumption (liters per 100 ASK or liters per 100 RPK<sup>23</sup>) and carbon intensity (g-  $CO_2/ASK$  and g- $CO_2/RPK$ ). By expressing measures in such metrics, we aim at making comparisons with previous literature (particularly with papers related to other transport modes). Carbon intensities are shown to differ widely between transport modes. In Europe, for example, the average consumption of a recent car is generally estimated to fall in the range of 5 to 10 liters per 100 km. To express it in liters per 100 ASK, we must divide this value by five (if we consider that five people may sit in a car). What concerns the train, the French train company SNCF estimates that the carbon intensity of the high-speed train TGV is equal to 22.3 g- $CO_2/ASK$ . This low value may be explained by the fact that electricity generation in France releases about 80g/kWh, because it relies heavily on nuclear power.  $CO_2$  emissions are assumed to be equal to 380g/kWh in the european electricity mix. Thus, we may estimate the carbon intensity of a TGV traveling in Europe to be equal to 105

 $<sup>^{21}</sup>$ The values in Table 1 are expressed not in ktoe per ATK but in gram of CO<sub>2</sub> per ASK and RPK. These mean values were computed during 2004-2006.

<sup>&</sup>lt;sup>22</sup>Hence, the carbon intensity computed with the 'macro-level' methodology hinges on an aircraft fleet aged between zero and thirty five years. Contrary to the 'Bottom-Up' methodology, we do not know exactly the composition of the aircraft fleet.

<sup>&</sup>lt;sup>23</sup>Revenue Passenger Kilometers (RPK) is a measure of the volume of passengers carried by an airline. A revenue passenger kilometer is flown when a revenue passenger is carried one kilometer.

			ASK		RPK		
Aircraft models	Release	Energy	Jet-Fuel	Carbon	Jet-Fuel	Carbon	
	date	efficiency	consumption	intensity	consumption	intensity	
		(MJ/ASK)	(1/100 ASK)	$(g CO_2/ASK)$	(1/100 RPK)	$(g CO_2/RPK)$	
		. , ,		(0 2/ /		(0 2/ /	
Comet 4	1958	3.500	10.16	248.66	13.19	322.93	
B707-120	1958	3.100	9.00	220.24	11.68	286.03	
B707-320	1959	3.020	8.77	214.56	11.38	278.65	
DC8-30	1961	2.975	8.63	211.36	11.21	274.49	
B707-120B	1962	2.520	7.31	179.03	9.50	232.51	
B707-320B	1963	2.205	6.40	156.65	8.31	203.45	
DC8-61	1965	2.310	6.70	164.11	8.71	213.14	
B737-200	1965	1.540	4.47	109.41	5.80	142.09	
SVC10	1966	2.380	6.91	169.09	8.97	219.59	
DC8-63	1967	1.890	5.49	134.28	7.12	174.38	
B747-100	1969	1.645	4.77	116.87	6.20	151.78	
B747-100B	1969	1.330	3.86	94.49	5.01	122.71	
B747-200	1970	1.680	4.88	119.36	6.33	155.01	
DC10-30	1972	1.715	4.98	121.84	6.46	158.24	
B747-200B	1972	1.330	3.86	94.49	5.01	122.71	
B747SP	1975	1.680	4.88	119.36	6.33	155.01	
A310-200	1982	1.256	3.65	89.25	4.73	115.91	
B767-200	1982	1.329	3.86	94.41	5.01	122.62	
B747-300	1983	1.225	3.56	87.03	4.62	113.03	
A300-600	1984	1.313	3.81	93.25	4.95	121.10	
A310-300	1985	1.120	3.25	79.57	4.22	103.34	
B737-400	1988	1.127	3.27	80.04	4.25	103.95	
A320	1988	1.018	2.96	72.35	3.84	93.96	
B747-400	1989	1.120	3.25	79.57	4.22	103.34	
A340-300	1993	1.155	3.35	82.06	4.35	106.57	
A330-300	1994	1.085	3.15	77.08	4.09	100.11	
B777-200	1994	1.050	3.05	74.60	3.96	96.88	
B777-300	1997	1.006	2.92	71.49	3.79	92.84	
B737-700	1997	0.893	2.59	63.46	3.37	82.41	
A380-800	2007	1.104	3.20	78.43	4.16	101.86	
B787-8 dreamliner	2010	1.063	3.09	75.53	4.01	98.09	
A350-900	2011	1.064	3.09	75.56	4.01	98.13	

Note: Aircrafts' energy efficiency (second column) have been obtained from carriers' websites. These data have then been converted into Jet-Fuel consumption (third and fifth columns), as well as in carbon intensity (fourth and sixth columns). The following assumptions have been made:

- PCI of Jet-A1: 44,23 MJ/kg;
- density of Jet-A1: 0,779 kg/l;
- carbon percentage (C) in Jet-A1: 85,7%;
- molecular weights of carbon dioxide  $(CO_2)$  and carbon (C): 44 and 12, respectively;
- aircraft's average filling rate to convert ASK into RPK: 77% (world average in 2007).

Table 4: Energy efficiency (expressed in Mega-Joule per ASK), Jet-Fuel consumption (expressed in liters per 100 ASK and 100 RPK) and carbon intensity (expressed in grammes of  $CO_2$  per ASK and RPK) of aircrafts. Source: Authors, based on manufacturers' data.

		Α	SK	RPK		
		Jet-Fuel	Carbon	Jet-Fuel	Carbon	
		consumption	intensity	consumption	intensity	
		(1/100 ASK)	$(g CO_2/ASK)$	(1/100 RPK)	$(g CO_2/RPK)$	
		()	(8 - 2/ - /	()	(8 - 2/ - )	
Central and	Aggregated	5.15	126.19	6.63	162.18	
North	Domestic	6.00	146.79	7.79	190.59	
America	International	3.60	88.01	4.58	112.00	
Europe	Aggregated	4.30	105.33	5.59	136.73	
	Domestic	9.86	241.43	14.50	354.95	
	International	3.81	93.19	4.89	119.63	
Latin	Aggregated	11.02	269.65	15.64	382.95	
America	Domestic	12.77	312.62	18.91	462.84	
	International	10.22	250.14	14.17	346.84	
Russia and	Aggregated	12.27	300.24	17.45	427.16	
CIS	Domestic	17.55	429.59	24.44	598.34	
	International	9.80	239.96	14.05	343.87	
Africa	Aggregated	9.23	225.98	13.64	333.90	
	Domestic	25.56	625.67	36.65	897.14	
	International	6.85	167.70	10.16	248.75	
The Middle	Aggregated	6.16	150.72	8.43	206.29	
East	Domestic	7.16	175.28	8.90	217.97	
	International	6.13	150.04	8.45	206.85	
Asian	Aggregated	4.91	120.14	6.78	166.04	
countries and	Domestic	5.79	141.80	8.47	207.33	
Oceania	International	4.74	116.10	6.50	159.10	
<i>c</i> 1.1						
China	Aggregated	4.07	99.72	5.54	135.66	
	Domestic	4.21	103.17	5.84	142.91	
	International	4.16	101.87	5.56	136.18	
World		5.14	125 70	6.91	166 69	
world	Aggregated	0.14 6.46	159.02	0.01	100.08	
	Domestic	0.40	107.60	8.04	211.47	
	International	4.40	107.69	5.80	142.02	

Note: These values were calculated using 'macro' methodology that aims at obtaining energy efficiency coefficients (expressed in ktoe/ASK or ktoe/RPK). These coefficients were then converted to jet-fuel consumption and carbon intensity. We used the following assumptions:

- 1 t of Jet-A1 = 1,061 to e of Jet-A1;
- Jet-A1 density: 0.779 kg/l;
- percentage of carbon (C) in Jet-A1: 85.7%;
- molecular weight of carbon dioxide  $(CO_2)$  and carbon (C): 44 and 12, respectively.

Table 5: Jet-Fuel consumption (expressed in liters per 100 ASK and 100 RPK) and carbon intensity (expressed in grammes of  $CO_2$  per ASK and RPK) of fleets within each zone, assessed on a global basis. Mean values for the years 2004-2006. Source: Authors, from ICAO and IEA data.

 $g-CO_2/ASK.$ 

It is generally assumed that air, railway and road are competitive for passengers transport up to 1,500km. For this range of distances, air transport appears as the less efficient transport mode from a strictly environmental viewpoint<sup>24</sup>. These comparisons between different transport modes reveal another interesting result. The carbon intensity of road transport is actually - especially for medium distances - lower than for rail. Indeed, the average carbon intensities of cars has decreased in most countries. The carbon intensity of new manufactured cars is comprised between 100 and 140 g-CO<sub>2</sub>/ km, which corresponds to 20 and 28 g-CO<sub>2</sub>/ASK respectively. It is lower than the carbon intensity of a TGV traveling in Europe (105 g-CO<sub>2</sub>/ASK), and close to the carbon intensity of a TGV traveling in France (22.3 g-CO<sub>2</sub>/ASK).

As illustrated by Table 5, the 'macro-level' methodology allows to compile the data at different levels of aggregation. Thus, we may estimate the carbon intensity of a representative aviation world's fleet (or aviation regional's fleet), for both domestic and international flights. Figure 6 shows the heterogeneity between carbon intensities at the regional level and depending on the types of flights.



Figure 6: Carbon intensity (expressed in  $g-CO_2/ASK$ ) of (*i*) aggregated, (*ii*) domestic and (*iii*) international air traffic by region in 2006. Source: Authors, from ICAO and IEA data.

The results shown in Table 5 are comparable with those in Tables 2 and 3. Two insights may be drawn from Table 5. First, the carbon intensity of the aviation fleet used for international flights (107 g-CO<sub>2</sub>/ASK, Table 3) is close to the carbon intensity of aircrafts that were

 $<sup>^{24}</sup>$ Unless we consider that short and medium hauls flight are opeated with A-380 models, which is not a realistic assumption.

recently manufactured (between 75 and 82 g-CO<sub>2</sub>/ASK, see Table 3). This result suggests that the fleet used for international flights is composed of relatively recent aircrafts. Second, when comparing the carbon intensities of North American and European regions, it appears that they are similar for international flights (112 g-CO<sub>2</sub>/RPK vs. 119 g-CO<sub>2</sub>/RPK), while they differ significantly for domestic flights (190 g-CO<sub>2</sub>/RPK vs. 355 g-CO<sub>2</sub>/RPK). This difference cannot be explained by a lower energy efficiency of the European companies, but it may be explained by the fact that European flights are shorter than North-American flights. Hence, the estimated value of the carbon intensity of domestic flights for these two regions does not correspond strictly to the same type of flights.

# 6 Concluding remarks

The 'Macro-Level' methodology presented in this article has been initially developed to estimate air transport energy efficiency. Compared to the 'Bottom-Up' methodology, the interest of using the 'Macro-Level' methodology lies in the fact that its results are i) all precisely quantified according to a simply replicable methodology, and ii) obtained without any (restrictive) assumptions on either the composition of the aircrafts' fleet or the evolution of the renewal/upgrade rate of existing aircrafts. Effectively, our results are obtained just by systematically comparing the evolution of both air traffic and Jet-Fuel consumptions among eight geographical zones during the last 30 years. More precisely, it allows us to obtain 'aggregated' (domestic + international), domestic and international EE coefficients and their growth rates from 1980 to 2006. These coefficients are provided at the world level and for eight geographical zones.

Our results indicate that, first, air travel energy efficiency improvements have been occurring in all regions, but not with the same magnitude (Table 1). At the world level, that is for the world aircraft fleet taken as a whole, energy efficiency improvements have been equal to 2.88% per year during the 1983-2006 period. Still at the world level, energy efficiency improvements have been more important during 1983-1996 (3.09% per year) than during 1996-2006 (2.61% per year). Second, it has been identified that some regions appear effectively more energy efficient than others (Table 2). Central and North America, Europe, China, Asia and Oceania are in average more energy efficient than the world's benchmark. Third, domestic energy efficiency appears to be lower than international energy efficiency. This latter comment applies both at the world level and for all regions (Table 3).

Taken together, these results highlight the necessity to take into account energy efficiency heterogeneity between aircraft fleets when estimating the energy efficiency of air transport. Two kinds of heterogeneities have to be distinguished. First, regional aircraft fleets do not have the same energy efficiency, as fleets are not composed with the same types of aircrafts. Second, domestic air traffic is less energy efficient than international air traffic, *ceteris paribus*. This is due to more frequent take-off and landing, the most energy-intensive components of a flight.

The heterogeneity between regions has important consequences in terms of forecasting  $CO_2$  emissions from air transport. For example, the rapid development of air transport will not have the same consequences on the overall increase in  $CO_2$  emissions in this sector if it occurs predominantly in relatively energy-efficient regions, *ceteris paribus*.

Similarly, the heterogeneity between different types of flights would encourage policymakers to ensure that the development of air traffic rely more on long-haul flights than on shortmedium haul flights. Air transport is indeed the less energy efficient transport mode from an environmental viewpoint when we consider the distances corresponding to the short- and medium-haul flights. Policy-makers should therefore promote other transportation means for distances inferior to 1,500 km.

Finally, we highlight that the quest for more energy efficient aircrafts has not been a priority for aircraft and engine manufacturers over the last thirty years. However, in the current economic outlook (i.e., increases in Jet-Fuel prices), greater energy efficiency gains are deemed as desirable. Last but not least, airline companies are now facing tight environmental constraints with the fight against climate change<sup>25</sup>. We may therefore expect further improvement of energy efficiency, and thus of carbon intensity of air transport in the coming years.

 $<sup>^{25}</sup>$ As airline companies transiting from and to Europe will fall under the regulation of the European Union Emissions Trading Scheme by 2012.

## References

Airbus, 2007. Flying by Nature: Global Market Forecast 2007–2026. Report. Airbus, France.

Bhadra, D., 2003. Demand for air travel in the United States: bottom-up econometric estimation and implications for forecasts by origin and destination pairs. Journal of Air Transportation 8(2), 19-56.

Bhadra, D., Kee, J. 2008. Structure and dynamics of the core US air travel markets: A basic empirical analysis of domestic passenger demand. Journal of Air Transport Management 14, 27-39.

Chèze, B., Gastineau, P., Chevallier, J., 2010. Forecasting Air Traffic and Corresponding Jet-Fuel Demand until 2025. Les cahiers de l'économie 77, Série Recherche, IFP School.

ECI, 2006. Predict and Decide: Aviation, Climate Change and UK Policy. Environmental Change Institute, Oxford.

Eyers, C., Norman, P., Middel, J., Plohr, M., Michot, S., Atkinson, K. and Christou, R., 2004. AERO2k Global Aviation Emissions Inventories for 2002 and 2025. QINETIQ Report to the European Commission, United Kingdom.

Frederick, J.H., 1961. Commercial air transportation. Richard D. Irwin, Inc. Homewood, Illinois.

Gately, D., 1988. Taking off: the US demand for air travel and jet fuel. The Energy Journal 9(2), 93-108.

Greene, D.L., 1992. Energy-efficiency improvement of commercial aircraft. Annual Review of Energy and the Environment 17, 537-573.

Greene, D.L., 1996. Transportation and Energy. Eno Transportation Foundation, Inc. Lansdowne, Va, USA, 1996.

Greene, D.L., 2004. Transportation and Energy, Overview. Encyclopedia of Energy 6, 179-188.

IEA, 2009a. CO<sub>2</sub> Emissions from Fuel Combustion. International Energy Agency, Paris.

IEA, 2009b. Transport, Energy and  $CO_2$ : Moving Towards Sustainability. International Energy Agency, Paris.

IEA, 2009c. World Energy Outlook. International Energy Agency, Paris.

IPCC, 1999. Aviation and the Global Atmosphere. in Intergovernmental Panel on Climate Change, Special Report of IPCC Working Group I and III, edited by Penner, J., Lister, D., Griggs, D., Dokken, D., and McFarland, M., Cambridge University Press, United Kingdom and United States.

IPCC, 2007a. Confronting Climate Change: Avoiding the Unmanageable and Managing the Unavoidable. in Intergovernmental Panel on Climate Change: A Special Report of IPCC Working Group I and III, 15th Session of the United Nations Commission on Sustainable Development, United Nations Foundation and The Scientific Research Society, USA.

IPCC, 2007b. Climate Change 2007: Summary for Policymaker of Synthesis Report. in Intergovernmental Panel on Climate Change, edited by Pachauri, R.K., and Reisinger, A., Cambridge University Press, Cambridge.

IPCC, 2007c. Climate Change 2007: Synthesis Report. in Intergovernmental Panel on Climate Change, edited by Pachauri, R.K., and Reisinger, A., Cambridge University Press, Cambridge.

Lee, J.J., Lukachko, S.P., Waitz, I.A. and Schafer, A., 2001. Historical and Future Trends in Aircraft Performance, Cost, and Emissions. Annual Review of Energy and the Environment 26, 167-200.

Lee, J.J., Lukachko, S.P., Waitz, I.A., 2004. Aircraft and energy use. Encyclopedia of Energy 1, 29-38.

Lee, D.S., Fahey, D.W., Forster, P.M., Newton, P.J., Wit, R.C.N, Lim, L.L., Owen, B., Sausen, R., 2009. Aviation and global climate change in the 21st century. Atmospheric Environment 43(22-23), 3520-3537.

Lee, J.J., 2010. Can we accelerate the improvement of energy efficiency in aircraft systems? Energy Conversion and Management 51, 189-196.

Macintosh, A., Wallace, L., 2009. International aviation emissions to 2025: Can emissions be stabilised without restricting demand? Energy Policy 37, 264-273.

Mayor, K., Tol, R.S.J., 2010. Scenarios of carbon dioxide emissions from aviation. Global Environmental Change 20, 65-73.

Mazraati M., Faquih Y.O., 2008. Modelling aviation fuel demand: the case of the United States and China. OPEC Energy Review, 323-342.

Owen, B., 2008. Fuel Efficiency Development and Prediction Main Thematic Area: Climate Change. MMU and Omega Working Paper.

Peeters, P.M., Middel, J., Hoolhorst, A., 2005. Fuel Efficiency of Commercial Aircraft -An Overview of Historical and Future Trends. NLR-CR-2005-669 Working Paper. National Aerospace Laboratory, Amsterdam.

RCEP, 2002. The Environmental Effects of Civil Aircraft in Flight. Royal Commission on Environmental Pollution, Government of the United Kingdom.

Vedantham, A., Oppenheimer, M., 1998. Long-term scenarios for aviation: Demand and

emissions of  $CO_2$  and  $NO_x$ . Energy Policy 26(8), 625-641.

# The "Cahiers de l'Économie" Series

The "*Cahiers de l'économie*" Series of occasional papers was launched in 1990 with the aim to enable scholars, researchers and practitioners to share important ideas with a broad audience of stakeholders including, academics, government departments, regulators, policy organisations and energy companies.

All these papers are available upon request at IFP School. All the papers issued after 2004 can be downloaded at: <u>www.ifpen.fr</u>

The list of issued occasional papers includes:

#### #1. D. PERRUCHET, J.-P. CUEILLE

Compagnies pétrolières internationales : intégration verticale et niveau de risque. Novembre 1990

#### # 2. C. BARRET, P. CHOLLET

Canadian gas exports: modeling a market in disequilibrium. Juin 1990

#### #3. J.-P. FAVENNEC, V. PREVOT

Raffinage et environnement. Janvier 1991

#### #4. D. BABUSIAUX

Note sur le choix des investissements en présence de rationnement du capital. Janvier 1991

#### # 5. J.-L. KARNIK

Les résultats financiers des sociétés de raffinage distribution en France 1978-89. Mars 1991

#### #6. I. CADORET, P. RENOU

Élasticités et substitutions énergétiques : difficultés méthodologiques. Avril 1991

#### #7. I. CADORET, J.-L. KARNIK

Modélisation de la demande de gaz naturel dans le secteur domestique : France, Italie, Royaume-Uni 1978-1989. Juillet 1991

#### #8. J.-M. BREUIL

Émissions de SO2 dans l'industrie française : une approche technico-économique. Septembre 1991

#### # 9. A. FAUVEAU, P. CHOLLET, F. LANTZ

Changements structurels dans un modèle économétrique de demande de carburant. Octobre 1991

#### # 10. P. RENOU

Modélisation des substitutions énergétiques dans les pays de l'OCDE. Décembre 1991

#### #11. E. DELAFOSSE

Marchés gaziers du Sud-Est asiatique : évolutions et enseignements. Juin 1992

#### #12. F. LANTZ, C. IOANNIDIS

Analysis of the French gasoline market since the deregulation of prices. Juillet 1992

#### # 13. K. FAID

Analysis of the American oil futures market. Décembre 1992

#### # 14. S. NACHET

La réglementation internationale pour la prévention et l'indemnisation des pollutions maritimes par les hydrocarbures. Mars 1993

#### # 15. J.-L. KARNIK, R. BAKER, D. PERRUCHET

Les compagnies pétrolières : 1973-1993, vingt ans après. Juillet 1993

#### #16. N. ALBA-SAUNAL

Environnement et élasticités de substitution dans l'industrie ; méthodes et interrogations pour l'avenir. Septembre 1993

#### # 17. E. DELAFOSSE

Pays en développement et enjeux gaziers : prendre en compte les contraintes d'accès aux ressources locales. Octobre 1993

#### #18. J.P. FAVENNEC, D. BABUSIAUX\*

L'industrie du raffinage dans le Golfe arabe, en Asie et en Europe : comparaison et interdépendance. Octobre 1993

#### # 19. S. FURLAN

L'apport de la théorie économique à la définition d'externalité. Juin 1994

#### # 20. M. CADREN

Analyse économétrique de l'intégration européenne des produits pétroliers : le marché du diesel en Allemagne et en France. Novembre 1994

#### # 21. J.L. KARNIK, J. MASSERON\*

L'impact du progrès technique sur l'industrie du pétrole. Janvier 1995

#### # 22. J.P. FAVENNEC, D. BABUSIAUX

L'avenir de l'industrie du raffinage. Janvier 1995

#### # 23. D. BABUSIAUX, S. YAFIL\*

Relations entre taux de rentabilité interne et taux de rendement comptable. Mai 1995

#### # 24. D. BABUSIAUX, J. JAYLET\*

Calculs de rentabilité et mode de financement des investissements, vers une nouvelle méthode ? Juin 1996

#### # 25. J.P. CUEILLE, J. MASSERON\*

Coûts de production des énergies fossiles : situation actuelle et perspectives. Juillet 1996

#### # 26. J.P. CUEILLE, E. JOURDAIN

Réductions des externalités : impacts du progrès technique et de l'amélioration de l'efficacité énergétique. Janvier 1997

#### # 27. J.P. CUEILLE, E. DOS SANTOS

Approche évolutionniste de la compétitivité des activités amont de la filière pétrolière dans une perspective de long terme. Février 1997

#### # 28. C. BAUDOUIN, J.P. FAVENNEC

Marges et perspectives du raffinage. Avril 1997

#### # 29. P. COUSSY, S. FURLAN, E. JOURDAIN, G. LANDRIEU, J.V. SPADARO, A. RABL

Tentative d'évaluation monétaire des coûts externes liés à la pollution automobile : difficultés méthodologiques et étude de cas. Février 1998

#### # 30. J.P. INDJEHAGOPIAN, F. LANTZ, V. SIMON

Dynamique des prix sur le marché des fiouls domestiques en Europe. Octobre 1998

#### # 31. A. PIERRU, A. MAURO

Actions et obligations : des options qui s'ignorent. Janvier 1999

#### # 32. V. LEPEZ, G. MANDONNET

Problèmes de robustesse dans l'estimation des réserves ultimes de pétrole conventionnel. Mars 1999

#### # 33. J. P. FAVENNEC, P. COPINSCHI

L'amont pétrolier en Afrique de l'Ouest, état des lieux Octobre 1999

#### # 34. D. BABUSIAUX

Mondialisation et formes de concurrence sur les grands marchés de matières premières énergétiques : le pétrole. Novembre 1999

#### # 35. D. RILEY

The Euro Février 2000

#### # 36. D. BABUSIAUX, A. PIERRU\*

Calculs de rentabilité et mode de financement des projets d'investissements : propositions méthodologiques. Avril 2000 & septembre 2000

#### # 37. P. ALBA, O. RECH

Peut-on améliorer les prévisions énergétiques ? Mai 2000

#### # 38. J.P. FAVENNEC, D. BABUSIAUX

Quel futur pour le prix du brut ? Septembre 2000

#### # 39. S. JUAN, F. LANTZ

La mise en œuvre des techniques de Bootstrap pour la prévision économétrique : application à l'industrie automobile Novembre 2000

#### # 40. A. PIERRU, D. BABUSIAUX

Coût du capital et étude de rentabilité d'investissement : une formulation unique de l'ensemble des méthodes. Novembre 2000

#### #41. D. BABUSIAUX

Les émissions de CO2 en raffinerie et leur affectation aux différents produits finis Décembre 2000

#### # 42. D. BABUSIAUX

Éléments pour l'analyse des évolutions des prix du brut. Décembre 2000

#### # 43. P. COPINSCHI

Stratégie des acteurs sur la scène pétrolière africaine (golfe de Guinée). Janvier 2001

#### # 44. V. LEPEZ

Modélisation de la distribution de la taille des champs d'un système pétrolier, Log Normale ou Fractale ? Une approche unificatrice. Janvier 2001

#### # 45. S. BARREAU

Innovations et stratégie de croissance externe : Le cas des entreprises parapétrolières. Juin 2001

#### # 46. J. P. CUEILLE\*

Les groupes pétroliers en 2000 : analyse de leur situation financière. Septembre 2001

#### #47. T. CAVATORTA

La libéralisation du secteur électrique de l'Union européenne et son impact sur la nouvelle organisation électrique française Décembre 2001

#### # 48. P. ALBA, O. RECH

Contribution à l'élaboration des scénarios énergétiques. Décembre 2001

#### # 49. A. PIERRU\*

Extension d'un théorème de dualité en programmation linéaire : Application à la décomposition de coûts marginaux de long terme. Avril 2002

#### # 50. T. CAVATORTA

La seconde phase de libéralisation des marchés du gaz de l'Union européenne : enjeux et risques pour le secteur gazier français. Novembre 2002

#### # 51. J.P. CUEILLE, L. DE CASTRO PINTO COUTHINO, J. F. DE MIGUEL RODRÍGUEZ\*

Les principales compagnies pétrolières indépendantes américaines : caractéristiques et résultats récents. Novembre 2002

#### # 52. J.P. FAVENNEC

Géopolitique du pétrole au début du XXI<sup>e</sup> siècle Janvier 2003

#### # 53. V. RODRIGUEZ-PADILLA avec la collaboration de T. CAVATORTA et J.P. FAVENNEC,\*

L'ouverture de l'exploration et de la production de gaz naturel au Mexique, libéralisme ou nationalisme Janvier 2003

#### # 54. T. CAVATORTA, M. SCHENCKERY

Les majors pétroliers vers le multi énergies : mythe ou réalité ? Juin 2003

#### # 55. P.R. BAUQUIS

What energy sources will power transport in the 21st century? Janvier 2004

#### # 56. A. PIERRU, D. BABUSIAUX

Evaluation de projets d'investissement par une firme multinationale : généralisation du concept de coût moyen pondéré du capital et conséquences sur la valeur de la firme. Février 2004

#### # 57. N. BRET-ROUZAUT, M. THOM

Technology Strategy in the Upstream Petroleum Supply Chain. Mars 2005

#### # 58. A. PIERRU

Allocating the  $CO_2$  emissions of an oil refinery with Aumann-Shapley prices. June 2005

#### # 59. F. LESCAROUX

The Economic Consequences of Rising Oil Prices. Mai 2006

#### # 60. F. LESCAROUX, O. RECH

L'origine des disparités de demande de carburant dans l'espace et le temps : l'effet de la saturation de l'équipement en automobiles sur l'élasticité revenu. Juin 2006

#### # 61. C. I. VASQUEZ JOSSE, A. NEUMANN

Transatlantic Natural Gas Price and Oil Price Relationships - An Empirical Analysis. Septembre 2006

#### # 62. E. HACHE

Une analyse de la stratégie des compagnies pétrolières internationales entre 1999 et 2004. Juillet 2006

#### #63. F. BERNARD, A. PRIEUR

Biofuel market and carbon modeling to evaluate French biofuel policy. Octobre 2006

#### # 64. E. HACHE

Que font les compagnies pétrolières internationales de leurs profits ? Janvier 2007

#### # 65. A. PIERRU

A note on the valuation of subsidized Loans Janvier 2007

#### # 66. D. BABUSIAUX, P. R. BAUQUIS

Depletion of Petroleum Reserves and Oil Price trends Septembre 2007

#### #67. F. LESCAROUX

Car ownership in relation to income distribution and consumers's spending decisions. Novembre 2007

#### # 68. D. BABUSIAUX, A. PIERRU

Short-run and long-run marginal costs of joint products in linear programming Juin 2008

#### # 69. E. HACHE

Commodities Markets: New paradigm or new fashion? Juillet 2008

#### # 70. D.BABUSIAUX, A. PIERRU

Investment project valuation: A new equity perspective Février 2009

#### #71. O. MASSOL, S. TCHUNG-MING

Stratégies coopératives dans l'industrie du GNL : l'argument de la rationalisation est-il fondé ? Février 2009

#### #72. A. PIERRU, D.BABUSIAUX

Valuation of investment projects by an international oil company: A new proof of a straightforward, rigorous method Février 2009

#### #73. E. SENTENAC CHEMIN

Is the price effect on fuel consumption symmetric? Some evidence from an empirical study Avril 2009

#### #74. E. HACHE

OBAMA : Vers un green New Deal énergétique ? Septembre 2009

#### # 75. O. MASSOL

Cost function for the natural gas transmission industry: further considerations Septembre 2009

#### # 76. F. LANTZ, E. SENTENAC CHEMIN

Analyse des tendances et des ruptures sur le marché automobile français. Modélisation du taux de diésélisation dans le parc Décembre 2010.

#### # 77. B. CHÈZE, P. GASTINEAU, J. CHEVALLIER

Forecasting air traffic and corresponding Jet-Fuel Demand until 2025 Décembre 2010.

#### #78. V. BREMOND, E. HACHE, V. MIGNON

Does OPEC still exist as a cartel? An empirical investigation Mars 2011.

#### #79. I. ABADA, O. MASSOL

Security of supply and retail competition in the European gas market. Some model-based insights. Mars 2011.

#### # 80. E. HACHE, F. LANTZ

Oil price volatility: an econometric analysis of the WTI market. Avril 2011.

#### # 81. I. ABADA, V. BRIAT, O. MASSOL

Construction of a fuel demand function portraying interfuel substitution, a system dynamics approach. Avril 2011

#### # 82. E. LE CADRE, F. LANTZ, P-A. JOUVET

The bioenergies development: the role of biofuels and the CO2 price. Décembre 2011

#### #83. E. LE CADRE, F. LANTZ, A. FARNOOSH

Bioenergies usages in electricity generation utility means through a modelling approach: application to the French case. Décembre 2011

#### # 84. I. ABADA, V. BRIAT, S. GABRIEL, O. MASSOL

A generalized Nash-Cournot model for the northwestern European natural gas markets with a fuel substitution demand function: the GaMMES model. Décembre 2011

#### # 85. O. MASSOL, A. BANAL-ESTAÑOL

Export diversification and resource-based industrialization: the case of natural gas. Décembre 2011

#### # 86. B. CHÈZE, P. GASTINEAU, J. CHEVALLIER

Air traffic energy efficiency differs from place to place: analysis of historical trends by geographical zones using a macro-level methodology

\* une version anglaise de cet article est disponible sur demande